Author's response to reviews

Title: An evidential reasoning based model for diagnosis of lymph node metastasis in gastric cancer

Authors:

Zhi-Guo Zhou (zgzhou@mail.xidian.edu.cn)
Fang Liu (F63liu@163.com)
Li-Cheng Jiao (Ichijiao@mail.xidian.edu.cn)
Zhi-Long Wang (zhilongwang.vip@gmail.com)
Xiao-Peng Zhang (zxp@bjcancer.org)
Xiao-Dong Wang (xdwang@mail.xidian.edu.cn)
Xiao-Zhuo Luo (etolxz@yahoo.com.cn)

Version: 5  Date: 24 October 2013

Author's response to reviews: see over
Reply to the reviewers’ comments

MS: 2048463939441054

Paper: An evidential reasoning based model for diagnosis of lymph node metastasis in gastric cancer

Authors: Zhi-Guo Zhou, Fang Liu, Li-Cheng Jiao, Zhi-Long Wang, Xiao-Peng Zhang, Xiao-Dong Wang and Xiao-Zhuo Luo

Journal: BMC Medical Informatics and Decision Making

Based on the comments from Editorial team and the reviewers, we have revised the paper carefully.

1. Reply to Editor-in-Chief

<table>
<thead>
<tr>
<th>Number</th>
<th>Comment</th>
<th>Response</th>
</tr>
</thead>
<tbody>
<tr>
<td>1</td>
<td>As you see, the reviewer considers you have successfully addressed their previous concerns. Before acceptance, we would ask to please carefully revise the new text included in this new version (parts in red), to improve the quality of written English. You may want to seek the assistance of a fluent English speaking colleague.</td>
<td>Thanks a lot. Agree and revised. We have revised the new text carefully and we also asked several colleagues who are native English speakers to check the English. We believe that the language is now acceptable for the review process.</td>
</tr>
</tbody>
</table>